Subject: Shift summary -- 3/16/16

From: Mark Sosebee <sosebee@uta.edu>

Date: 03/30/2016 10:37 AM

To: Mark Sosebee <sosebee@uta.edu>

Shift notes from the past week:

ADCoS/CRC reports from the ADC Weekly and ADCoS meetings: https://indico.cern.ch/event/493009/contribution/2/attachments/1243858/1830737 /CRC Report 150316.pdf (CRC report)

- 1) 3/13: OU_OCHEP_SWT2 squid service down. Issue was eventually traced to a networking/LHCONE problem. Problem was fixed as of 3/15, so https://qqus.eu/?mode=ticket_info&ticket_id=120155 was closed. (Also some ddm deletion errors during this period due to the same problem.) eLog 56746.
- 2) 3/13: Large numbers of file transfer failures with "no FTS server had updated the transfer status the last 600 seconds, probably stalled in fts301.usatlas.bnl.gov." Cleaned some space filled up by log files issue resolved. https://ggus.eu/index.php?mode=ticket info&ticket id=120156 was closed, eLog 56783.
- 3) 3/14: New pilot release from Paul (v64.5): http://www-hep.uta.edu/~sosebee/ADCoS/pilot-release-v64 5-3 14 16.pdf
- 4) 3/15: ADC Weekly meeting: https://indico.cern.ch/e/493009

Follow-ups from earlier reports:

None

1 of 2 03/30/2016 10:42 AM

Shift summary -- 3/16/16

2 of 2 03/30/2016 10:42 AM